**Author's response to reviews**

**Title:** Epstein Barr Virus (EBV)-positive large T-cell lymphoma, 17 years after cadaveric renal transplant presenting as acute appendicitis

**Authors:**

Shiva K Ratuapli (shiva.ratuapli@bannerhealth.com)
Shishir Murarka (shishir.murarka@bannerhealth.com)
Karen Miller (karen.miller@bannerhealth.com)
James Ferraro (james.ferraro@bannerhealth.com)
Haider Zafar (thezafs@aol.com)

**Version:** 2  **Date:** 15 October 2010

**Author's response to reviews:** see over
To  
The Editor,  
Journal of Medical Case Reports.  

Dear Sir,  

I am submitting a revised version of our manuscript titled “Epstein Barr Virus (EBV) positive, Large T-Cell Lymphoma, 17 years after Cadaveric Renal Transplant Presenting as Acute Appendicitis”, in the case report category of your journal. We used one of the professional copy editing services suggested by you, to copy – edit our manuscript. We believe this case is unusual and merits publishing for these reasons.

- This is a relatively rare subtype (T cell type) of Post Transplant Lymphoproliferative Disorders.
- This case has unusual initial presentation as acute appendicitis, which is rarely reported.
- Despite T-cell lineage this case was associated with EBV positivity, which is rarely seen in such cases.
- Successful treatment outcome in light very high reported mortality provides a good report and literature review for practicing physicians.

Based on these findings we believe this is a strong case to report. We also affirm that this is the original work of the authors, and is not under simultaneous review by any other journal. We now request you to review our revised version of the manuscript and consider for publication.

Thank You,  

Sincerely,  

Shiva Ratuapli, MD.